Author's response to reviews

Title: Communication and Proximity Affects Attributable to Sense of Presence in Distance Bioinformatics Education

Authors:

Craig Locatis (locatis@nlm.nih.gov)
Eta Berner (bernere@uab.edu)
Glenn Hammack (glenn.hammack@nuphysicia.com)
Steve Smith (ssmith@uabsom.uab.edu)
Richard Maisiak (rmaisiak@yahoo.com)
Michael Ackerman (ackerman@nlm.nih.gov)

Version: 5 Date: 1 February 2011

Author's response to reviews: see over
Response to suggestions from the Associate Editor

We reviewed the manuscript for grammatical errors and corrected those detected. The word processing software was set to the strictest level of grammatical error detection and we attempted to identify and edit any possibly ambiguous prose.

We edited the first paragraph of the discussion section and deleted the second to the last sentence in the conclusion.

We added sentences in the introductory section about the study’s significance and intent and in the conclusion section about its implications.